\begin{thebibliography}{10}\setlength{\itemsep}{-1ex}\small

\bibitem{Bai_ITSC:06}
F.~Bai and H.~Krishnan.
\newblock {Reliability Analysis of DSRC Wireless Communication for Vehicle
  Safety Applications}.
\newblock In {\em IEEE Intelligent Transportation Systems Conference
  (ITSC'06)}, pages 355--362, Toronto, Canada, Sept. 2006.

\bibitem{Bilstrup_VTC:08}
K.~Bilstrup, E.~Uhlemann, E.~G. Str\"{o}m, and U.~Bilstrup.
\newblock {Evaluation of the IEEE 802.11p MAC method for Vehicle-to-Vehicle
  Communication}.
\newblock In {\em IEEE 68th Vehicular Technology Conference (VTC '08-Fall)},
  pages 1--5, Calgary, Alberta, Canada, Sept. 2008.

\bibitem{Cahill_WOWMOM:06}
M.~Bouroche, B.~Hughes, and V.~Cahill.
\newblock {Building Reliable Mobile Applications with Space-Elastic
  Adaptation}.
\newblock In {\em Proceedings of the 2006 International Symposium on on World
  of Wireless, Mobile and Multimedia Networks (WOWMOM '06)}, pages 627--632,
  Niagara Falls, NY, USA, 2006. IEEE Computer Society.

\bibitem{Bui_RTAS:07}
B.~D. Bui, R.~Pellizzoni, M.~Caccamo, C.~F. Cheah, and A.~Tzakis.
\newblock {Soft Real-Time Chains for Multi-Hop Wireless Ad-Hoc Networks}.
\newblock In {\em 13th IEEE Real Time and Embedded Technology and Applications
  Symposium (RTAS'07)}, pages 69--80, Seattle, WA, 2007.

\bibitem{Burgess_MaxProp:06}
J.~Burgess, B.~Gallagher, D.~Jensen, and B.~N. Levine.
\newblock {MaxProp: Routing for Vehicle-Based Disruption-Tolerant Networks}.
\newblock In {\em Proceedings of the 25th IEEE International Conference on
  Computer Communications (Infocom '06)}, pages 1--11, Barcelona, Spain, 2006.
  IEEE Comm Soc.

\bibitem{Chandra_HOTMOBILE:07}
R.~Chandra, J.~Padhye, L.~Ravindranath, and A.~Wolman.
\newblock {Beacon-Stuffing: Wi-Fi without Associations}.
\newblock In {\em Proceedings of the Eighth IEEE Workshop on Mobile Computing
  Systems and Applications (HOTMOBILE '07)}, pages 53--57, Tucson, AZ, USA,
  2007. IEEE Computer Society.

\bibitem{Chen_MW:03}
Y.~Chen, K.~Schwan, and D.~Zhou.
\newblock {Opportunistic Channels: Mobility-aware Event Delivery}.
\newblock In {\em Proceedings of the ACM/IFIP/USENIX 2003 International
  Conference on Middleware (Middleware '03)}, pages 182--201, Rio de Janeiro,
  Brazil, 2003. Springer-Verlag New York, Inc.

\bibitem{Eichler_VTC:07}
S.~Eichler.
\newblock {Performance Evaluation of the IEEE 802.11 p WAVE Communication
  Standard}.
\newblock In {\em IEEE 66th Vehicular Technology Conference (VTC Fall '07)},
  pages 2199--2203, Baltimore, MD, Oct. 2007.

\bibitem{Friis_Xmit:46}
H.~T. Friis.
\newblock {A Note on a Simple Transmission Formula}.
\newblock {\em proc. IRE}, 34(5):254--256, 1946.

\bibitem{Jain_DTN:04}
S.~Jain, K.~Fall, and R.~Patra.
\newblock {Routing in a Delay Tolerant Network}.
\newblock In {\em Proceedings of the 2004 Conference on Applications,
  Technologies, Architectures, and Protocols for Computer Communications
  (SIGCOMM '04)}, pages 145--158, Portland, Oregon, USA, 2004. ACM.

\bibitem{VTC_80211p:08}
D.~Jiang and L.~Delgrossi.
\newblock {IEEE 802.11p: Towards an International Standard for Wireless Access
  in Vehicular Environments}.
\newblock In {\em IEEE Vehicular Technology Conference (VTC '08)}, pages
  2036--2040, May 2008.

\bibitem{Jiang_DSRC:06}
D.~Jiang, V.~Taliwal, A.~Meier, W.~Holfelder, and R.~Herrtwich.
\newblock {Design of 5.9 Ghz DSRC-based Vehicular Safety Communication}.
\newblock {\em IEEE Wireless Communications}, 13(5):36--43, Oct. 2006.

\bibitem{ComPoScan_MobiSys:08}
T.~King and M.~B. Kj{\'a}ergaard.
\newblock {ComPoScan: Adaptive Scanning for Efficient Concurrent Communications
  and Positioning with 802.11}.
\newblock In {\em 6th International ACM Conference on Mobile Systems,
  applications, and services (MobiSys '08)}, pages 67--80, Breckenridge, CO,
  USA, 2008. ACM.

\bibitem{Perkins_MobileIP:97}
C.~E. Perkins.
\newblock {Mobile IP}.
\newblock {\em IEEE Communications}, 35(5):84--99, 1997.

\bibitem{Rudack_IVC:02}
M.~Rudack, M.~Meincke, and M.~Lott.
\newblock {On the Dyanamics of Ad Hoc Networks for Inter Vehicle Communications
  (IVC)}.
\newblock In {\em Proceedings of the International Conference on Wireless
  Networks (ICWN '02)}, Las Vegas, NV, USA, 2002.

\bibitem{Wang_ISCC:08}
S.-Y. Wang, H.-L. Chao, K.-C. Liu, T.-W. He, C.-C. Lin, and C.-L. Chou.
\newblock {Evaluating and Improving the TCP/UDP Performances of IEEE
  802.11(p)/1609 Networks}.
\newblock In {\em IEEE Symposium on Computers and Communications (ISCC '08)},
  pages 163--168, Marrakech, Morocco, July 2008.

\bibitem{Wang_ICVES:08}
Y.~Wang, A.~Ahmed, B.~Krishnamachari, and K.~Psounis.
\newblock {IEEE 802.11p Performance Evaluation and Protocol Enhancement}.
\newblock In {\em IEEE International Conference on Vehicular Electronics and
  Safety (ICVES '08)}, pages 317--322, Columbus, OH, Sept. 2008.

\bibitem{Weiland_ITS:00}
R.~J. Weiland and L.~B. Purser.
\newblock {Intelligent Transportation Systems}.
\newblock {\em Transportation Research}, 1:40AM, 2009.

\bibitem{Wellens_VTC:07}
M.~Wellens, B.~Westphal, and P.~Mahonen.
\newblock {Performance Evaluation of IEEE 802.11-based WLANs in Vehicular
  Scenarios}.
\newblock In {\em IEEE 65th Vehicular Technology Conference (VTC '07-Spring)},
  pages 1167--1171, Dublin, Ireland, Apr. 2007.

\bibitem{Wolf_CPS:09}
W.~Wolf.
\newblock Cyber-physical systems.
\newblock {\em IEEE Computer}, 42(3):88--89, 2009.

\end{thebibliography}

% $Id: relwork.tex 20822 2010-01-03 14:28:25Z gokhale $
\section{Related Work}
\label{sec:relwork}

Prior research closely related to the ideas presented in this paper
can be classified along two dimensions: (1) those that evaluate the
performance of the wireless channel for vehicular communications and
(2) those that develop higher-level capabilities (\emph{e.g.}, in
middleware) to overcome the challenges imposed by mobility.  These
prior efforts are important since they provide crucial insights into
the behavior and performance of the physical medium, or provide key
software-based algorithms that can tune performance.

\subsection{Research on Performance Evaluation of Wireless Networks in the Presence of Mobility} 

The IEEE 802.11 technologies form the basis of most wireless networks
in wireless LANs and Wi-Fi, which have become affordable and
commoditized.  The vehicular networks are also designed to leverage
these dominant technologies.  In particular, the IEEE
802.11p~\cite{Jiang_DSRC:06,VTC_80211p:08} is an emerging standard for
dedicated short range communication (DSRC) used in vehicular networks.

Bai~\cite{Bai_ITSC:06} describes real-world experiments involving
three vehicles.  The authors focus primarily on the reliability aspect
of 802.11p DSRC for vehicular safety communications needed in
applications, such as forward collision warning or blind spot
advising.  The authors measured the packet delivery ratio
(\emph{i.e.}, the percentage of packets transmitted that were
correctly received) and distribution of consecutive packet drops
(\emph{i.e.}, the probability distribution of consecutive packet
drops).  The authors concluded that DSRC does not incur packet loss in
bursts and hence the reliability of DSRC is adequate.

Despite their critical findings, the insights in~\cite{Bai_ITSC:06}
are limited due to measurements with only three vehicles using only
V2V communications.  Vehicular networks simultaneously must
support both the V2V and V2I communication networks, and that too with
variation in density of vehicles, speeds, and direction of traffic
flow.

Prior
studies~\cite{Bilstrup_VTC:08,Eichler_VTC:07,Wang_ISCC:08,Wang_ICVES:08,Wellens_VTC:07}
have evaluated the performance of 802.11p using simulations and
analytical models.  Eichler~\cite{Eichler_VTC:07} conducted
OMNeT++-based simulations to measure the collision probability,
throughput, and delay for 802.11p.  The study determined that 802.11p
was unable to handle many high priority messages in a dense network of
vehicles.  Similarly, lower priority messages suffer from exponential
delays as the density of the network increases.

Bilstrup et. al.~\cite{Bilstrup_VTC:08} 
%~\cite{Bilstrup_ITS:08,Bilstrup_VTC:08} 
study 802.11p MAC layer for supporting real-time
applications.  In contrast to the findings in~\cite{Bai_ITSC:06}, they
conclude that the channel access (which is a property controlled by
the MAC layer) cannot be assured in a manner that supports predictable
behavior.  As a result, 802.11p is not suitable for low-latency
communications.

Similarly, Wang et. al~\cite{Wang_ICVES:08} conclude that certain
parameter settings in 802.11p MAC can lead to undesirable throughput.
For example, the backoff window sizes (which are needed after
collisions) are not adaptive to the dynamics of the number of vehicles
involved in the communication.  This problem stems from the density of
vehicles and the percentage of them trying to communicate.

All these findings are crucial since solutions to overcome the
inherent limitations are needed.  In~\cite{Wang_ISCC:08}, the authors
modify the 802.11p MAC layer to avoid wastage of bandwith resulting
from switching of channels between 802.11p's control channel and
service channel.  This results in better observed performance at the
TCP/UDP level.  Similarly, in~\cite{Bui_RTAS:07}, the authors modify
the MAC layer to alleviate the problems of collisions and priority
inversions thereby supporting real-time communications.  Although the
solutions in our paper also overcome similar challenges, we focus on
integrating individual solutions at the both the physical and cyber
level.

\subsection{Research on Application Layer Software Capabilities}
The JECho event infrastructure research~\cite{Chen_MW:03} argues that
a static topology of brokers is insufficient to assure reliable
delivery of information when end hosts are mobile.  To overcome this
problem, the authors present an approach called \emph{opportunistic
event channels,} in which brokers are dynamically created or deleted
depending on the mobility of the producers and consumers.  An
algorithm to optimize the event delivery path is presented that
considers the network-level handoffs to kickstart an application-level
handoff of the producer or consumer from its broker to a new broker
that can provide higher assurances of data delivery. 

The RT-STEAM~\cite{Cahill_WOWMOM:06} project describes a space-elastic
model to provide real-time information dissemination.  The model is
elastic because the range of dissemination can be adapted to continue
to meet real-time requirements.  The approach is spatial because it
relies on geographical proximity of interested subscribers from the
producers to adapt the region of dissemination.  This approach is
geared towards ad hoc communications between vehicles.

The ComPoScan~\cite{ComPoScan_MobiSys:08} project trades off the
available window for data communications with positioning accuracy by
switching between active scanning of signal-to-noise ratio versus
monitor sniffing, which is supported on majority of 802.11 cards.
This work is important since it can form a building block for a more
holistic ITS capability.

Despite significant prior work, however, these efforts tend to focus
on individual problems addressing them in isolation.  Our focus is to
address the challenges holistically as a CPS.
% Andy, you might briefly explain how the rest of the paper addresses
% these challenges.
% DONE
Our CPS solution relies on sensing physical parameters, such as
vehicle speed and signal-to-noise ratio, to trigger the runtime
adaptation logic supported by the cyber layer, which in turn controls
physical parameters, such as wireless radio power.

%%% Local Variables: 
%%% mode: latex
%%% TeX-master: "iccps2010"
%%% End: 
